

Author's response to reviews

Title: WEARCON: Wearable home monitoring in children with asthma reveals a strong association with hospital based assessment of asthma control.

Authors:

Mattienne van der Kamp (m.r.vanderkamp-1@utwente.nl;mattienne@gmail.com;m.vanderkamp@mst.nl)

Emilie C Klaver (e.klaver@mst.nl)

Bernard J Thio (b.thio@mst.nl)

Jean MM Driessen (j.driessen@ocon.nl)

Franciscus HC de Jongh (f.h.c.dejongh@utwente.nl)

Monique Tabak (m.tabak@rrd.nl)

Job van der Palen (j.vanderpalen@mst.nl)

Hermie J Hermens (h.hermens@utwente.nl)

Version: 3 **Date:** 30 Jul 2020

Author's response to reviews:

1. The old files are removed. The proofread manuscript and the separate figure files are left.
2. The declarations are moved to follow the abbreviations
3. No copyright required.
4. All tracked changes etc are removed and figure/table citations are checked.